Author’s response to reviews

Title: How school nurses experience and understand everyday pain among adolescents

Authors:

Magnhild Høie (magnhild.hoie@ui.no)
Kristin Haraldstad (kristin.haraldstad@ui.no)
Gudrun Rohde (gudrun.e.rohde@ui.no)
Liv Fegran (liv.fegran@ui.no)
Thomas Westergren (thomas.westergren@ui.no)
Sølvi Helseth (solvi.helseth@hioa.no)
Åshild Slettebø (ashild.slettebo@ui.no)
Berit Johannessen (berit.johannessen@ui.no)

Version: 2 Date: 03 Aug 2017

Author’s response to reviews:

Magnhild Høie
University of Agder
Faculty of health and sport sciences
Postbox 422
4604 Kristiansand, Norway

Dear Editor,

I hereby submit our revised manuscript R2:“ "How school nurses experience and understand everyday pain among adolescents" (NURS-D-16-00128)” Magnhild Høie, Kristin Haraldstad, Gudrun Rohde, Liv Fegran, Thomas Westergren, Sølvi Helseth, Åshild Slettebø, Berit Johannessen

Arendal 02.08.2017
I hope we have addressed the editor’s comments properly. If there still is something missing in our manuscript or submission, please let us know. The revised manuscript has been reviewed and approved by my co-authors.

Yours sincerely

Magnhild Høie, PhD,
Associate Professor

Reply to editor’s comments:

We have copied the comments into the document and our replies are in italic:

1. Availability of Data and Materials (ADM) statement:

The statement "No further data are available" is not valid.

For your information: Here are some examples of Availability of Data and Materials statements that you may choose to adapt if you wish:

• The datasets generated during and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK TO DATASETS]

• The datasets during and/or analysed during the current study available from the corresponding author on reasonable request.

• The datasets generated during and/or analysed during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC] but are available from the corresponding author on reasonable request.

• The datasets generated during the current study are not publicly available due [REASON WHY DATA ARE NOT PUBLIC].

Reply: We have made the changes in accordance with the editor’s suggestion and written the sentence: “The dataset analysed during the current study is available from the corresponding author on reasonable request”.

2. Tracked changes

Please upload a single final, clean version of the manuscript with tracked changes switched off (and please ignore any automated requests to show tracked changes).

Reply: We have uploaded a single final, clean version of the manuscript